\begin{table}%t2
\caption{\label{tab:photometry_fir} Mid-infrared to radio photometry of SHADES galaxies in the Lockman hole east.}
\par
%\centering
\par
\begin{tabular}{llllllllll llllllllll llllllllll }
\hline\hline \noalign{\smallskip}
Name        & $15~\mu$m ($\mu$Jy)  & $24~\mu$m  ($\mu$Jy) & $350~\mu$m (mJy) & $850~\mu$m (mJy) & $1.1$~mm (mJy) & $1.2$~mm (mJy) & $1.4~$GHz  ($\mu$Jy) \\
\hline\noalign{\smallskip}
Lock850.001 & 25 $\pm$ 20   &217 $\pm$ 16&24.1 $\pm$ 5.5& $8.85^{\rm +1.0}_{-1.0}$ & $4.65_{-0.98}^{\rm +0.97}$ &3.6 $\pm$ 0.5 &78.9 $\pm$ 4.7\\
Lock850.002 & 13 $\pm$ 20   &545 $\pm$ 31&25.3 $\pm$ 10.3& $13.45^{\rm +2.1}_{-2.1}$ & $6.21_{-0.92}^{\rm +1.05}$ &5.7 $\pm$ 1.0 &52.4 $\pm$ 5.6\\
Lock850.003 & 208 $\pm$ 20  &175 $\pm$ 23&40.5 $\pm$ 6.5& $10.95^{\rm +1.8}_{-1.9}$ & $3.01_{-1.06}^{\rm +1.04}$ &4.6 $\pm$ 0.4 &25.8 $\pm$ 4.9\\
Lock850.003b& 438 $\pm$ 20  &183 $\pm$ 33&40.5 $\pm$ 6.5& $10.95^{\rm +1.8}_{-1.9}$ & $3.01_{-1.06}^{\rm +1.04}$ &4.6 $\pm$ 0.4 &35.0 $\pm$ 5.2\\
Lock850.004 & 118 $\pm$ 19  &179 $\pm$ 68&24.9 $\pm$ 9.1& $10.65^{\rm +1.7}_{-1.8}$ & $4.75_{-1.06}^{\rm +0.89}$ &3.7 $\pm$ 0.4 &32.0 $\pm$ 5.1\\
Lock850.004b& 59 $\pm$ 19   &261 $\pm$ 73&& $10.65^{\rm +1.7}_{-1.8}$ & 4.75$_{-1.06}^{\rm +0.89}$ & &73.0 $\pm$ 5.0\\
Lock850.005 & 770 $\pm$ 20  &58.6 $\pm$ 15.1&& $8.15^{\rm +2.0}_{-2.1}$ && &$<$22\\
Lock850.006 & 94 $\pm$ 20   &75.1 $\pm$ 12.7&38.0 $\pm$ 37.6& $6.85^{\rm +1.3}_{-1.3}$ & $2.13_{-1.26}^{\rm +1.12}$ & &15 $\pm$ 4.8\\
Lock850.007 & 35 $\pm$ 19   &341 $\pm$ 21&& $8.55^{\rm +1.8}_{-1.9}$ && &42.6 $\pm$ 5.8\\
Lock850.008 &             &481 $\pm$ 25&& $5.45^{\rm +1.1}_{-1.2}$ && &$<$22\\
Lock850.009 & 87 $\pm$ 20   &466 $\pm$ 74&& $5.95^{\rm +1.6}_{-1.6}$ && &52.6 $\pm$ 4.7\\
Lock850.009b& 87 $\pm$ 20   &159 $\pm$ 73&& $5.95^{\rm +1.6}_{-1.6}$ & & \\
Lock850.010 &             &
&24.1 $\pm$ 12.0& $9.15^{\rm +2.7}_{-2.9}$ & & &25.5 $\pm$ 6.3\\
Lock850.010b&             &79.6 $\pm$ 10.8& & $9.15^{\rm +2.7}_{-2.9}$ & & &
\\
Lock850.011 & 105 $\pm$ 20  &112 $\pm$ 57&& $6.25^{\rm +1.7}_{-1.8}$ & & &
\\
Lock850.012 & 9 $\pm$ 19    &263 $\pm$ 19&29.3 $\pm$ 16.0& $6.15^{\rm +1.7}_{-1.7}$ &4.1 $\pm$ 1.3&2.6 $\pm$ 0.4 &44.3 $\pm$ 5.1\\
Lock850.013 & 47 $\pm$ 20   &172 $\pm$ 14&& $5.65^{\rm +2.3}_{-2.9}$ & 3.07$_{-0.97}^{\rm +0.91}$ & &$<$28\\
Lock850.014 & --45 $\pm$ 20  &188 $\pm$ 16&41.0 $\pm$ 6.8& $7.25^{\rm +1.8}_{-1.9}$ & $1.79_{-1.57}^{\rm +0.97}$ &3.4 $\pm$ 0.6 &37.4 $\pm$ 4.2\\
Lock850.015 & 299 $\pm$ 19  &353 $\pm$ 20&6.5 $\pm$ 55.7& $13.25^{\rm +4.3}_{-5.0}$ & $3.61_{-0.98}^{\rm +1.02}$ &4.1 $\pm$ 0.7 &61.5 $\pm$ 7.6\\
Lock850.015b& 299 $\pm$ 19  &70.4 $\pm$ 12.1&6.5 $\pm$ 55.7& $13.25^{\rm +4.3}_{-5.0}$ & $3.61_{-0.98}^{\rm +1.02}$ &4.1 $\pm$ 0.7 &43.9 $\pm$ 7.8\\
Lock850.016 & 321 $\pm$ 19  &314 $\pm$ 24&25.7 $\pm$ 15.8& $5.85^{\rm +1.8}_{-1.9}$ &&1.8 $\pm$ 0.5 &106 $\pm$ 6\\
Lock850.017 &             &239 $\pm$ 18&& $4.75^{\rm +1.3}_{-1.3}$ && &92.3 $\pm$ 4.5\\
Lock850.017b&             &64.2 $\pm$ 26.1&& $4.75^{\rm +1.3}_{-1.3}$ && &Confused\\
Lock850.018 & 40 $\pm$ 19   &
&7.5 $\pm$ 6.7& $6.05^{\rm +1.9}_{-2.1}$ &5.1 $\pm$ 1.3&3.4 $\pm$ 0.6 &29.4 $\pm$ 4.4\\
Lock850.019 &             &118 $\pm$ 15&& $5.15^{\rm +2.0}_{-2.4}$ && &$<$27\\
Lock850.021 &             &97.9 $\pm$ 14.1&16.7 $\pm$ 13.2& $4.15^{\rm +2.0}_{-2.5}$ &&1.6 $\pm$ 0.4 &$<$30\\
Lock850.022 & --15 $\pm$ 20  &402 $\pm$ 21&8.7 $\pm$ 15.6& $7.55^{\rm +3.2}_{-4.2}$ && &$<$30\\
Lock850.023 &             &
&& $4.35^{\rm +1.9}_{-2.4}$ && &$<$25\\
Lock850.024 & --1 $\pm$ 30   &455 $\pm$ 21&& $2.75^{\rm +1.2}_{-1.2}$ && &28.5 $\pm$ 4.8\\
Lock850.026 & --80 $\pm$ 28  &195 $\pm$ 16&12.2 $\pm$ 8.8& $5.85^{\rm +2.4}_{-2.9}$ && &31.4 $\pm$ 5.2\\
Lock850.027 &             &106 $\pm$ 15&3.4 $\pm$ 5.1& $5.05^{\rm +1.3}_{-1.3}$ &5.2 $\pm$ 1.4&3.2 $\pm$ 0.7 &
\\
Lock850.028 &             &252 $\pm$ 14&23.3 $\pm$ 11.7& $6.45^{\rm +1.7}_{-1.8}$ && &63.0 $\pm$ 8.2\\
Lock850.029 &             &111 $\pm$ 14&& $6.75^{\rm +2.0}_{-2.2}$ && &23.7 $\pm$ 4.9\\
Lock850.030 &             &233 $\pm$ 19&38.0 $\pm$ 7.2& $4.75^{\rm +1.5}_{-1.6}$ &&0.4 $\pm$ 0.8 &245 $\pm$ 13\\
Lock850.031 &             &467 $\pm$ 19&& $6.05^{\rm +1.8}_{-2.0}$ && &43.0 $\pm$ 4.7\\
Lock850.033 & --71 $\pm$ 20  &104 $\pm$ 14&16.5 $\pm$ 8.4& $3.85^{\rm +1.0}_{-1.1}$ &&2.8 $\pm$ 0.6 &51.0 $\pm$ 4.3\\
Lock850.034 &             &84.9 $\pm$ 16.7&& $14.05^{\rm +3.1}_{-3.2}$ & 4.09$_{-0.92}^{\rm +0.90}$ & &58.4 $\pm$ 8.5\\
Lock850.035 & --25 $\pm$ 20  &51.0 $\pm$ 12.7&& $6.15^{\rm +2.2}_{-2.4}$ && &17.4 $\pm$ 5.0\\
Lock850.035b& --53 $\pm$ 20  &161 $\pm$ 14&& $6.15^{\rm +2.2}_{-2.4}$ && &
\\
Lock850.036 &             &$<$60&& $6.35^{\rm +1.7}_{-1.8}$ && &$<$20\\
Lock850.037 &             &
&& $7.55^{\rm +2.9}_{-3.5}$ && &41.8 $\pm$ 8.7\\ 
Lock850.037b&             &250 $\pm$ 17&& $7.55^{\rm +2.9}_{-3.5}$ && &14.8 $\pm$ 5.4\\[2pt]
\hline
Lock850.038 & 218 $\pm$ 19  &260 $\pm$ 16&& $4.35^{\rm +2.2}_{-2.7}$ && &24.4 $\pm$ 6.7\\
Lock850.039 &             &$<$60&& $6.55^{\rm +2.2}_{-2.7}$ && &$<$20\\
Lock850.040 &             &91.9 $\pm$ 15.0&& $3.05^{\rm +1.1}_{-1.2}$ && &16.2 $\pm$ 4.3\\
Lock850.041b& 193 $\pm$ 20  &651 $\pm$ 46&10.3 $\pm$ 5.5& $3.85^{\rm +0.9}_{-1.0}$ &4.0 $\pm$ 1.3&2.4 $\pm$ 0.5 &22.1 $\pm$ 4.8\\
Lock850.041 & 464 $\pm$ 20  &475 $\pm$ 37&10.3 $\pm$ 5.5& $3.85^{\rm +0.9}_{-1.0}$ &4.0 $\pm$ 1.3&2.4 $\pm$ 0.5 &43.6 $\pm$ 4.7\\
Lock850.043 & 140 $\pm$ 19  &261 $\pm$ 24&& $4.95^{\rm +2.1}_{-2.6}$ & 3.19$_{-0.99}^{\rm +1.10}$ & &25.4 $\pm$ 5.4\\
Lock850.043b& 267 $\pm$ 19  &456 $\pm$ 35&& $4.95^{\rm +2.1}_{-2.6}$ & 3.19$_{-0.99}^{\rm +1.10}$ & &40.8 $\pm$ 5.9\\
Lock850.047 & 37 $\pm$ 20   &107 $\pm$ 16&16.3 $\pm$ 20.9& $3.55^{\rm +1.7}_{-2.1}$ && &$<$22\\
Lock850.048 &             &203 $\pm$ 17&16.2 $\pm$ 13.5& $5.45^{\rm +2.1}_{-2.5}$ &&1.6 $\pm$ 0.4 &43.7 $\pm$ 10.0\\
Lock850.052 &             &310 $\pm$ 35&& $3.95^{\rm +2.2}_{-2.7}$ & 2.51$_{-1.02}^{\rm +1.03}$ & &38.7 $\pm$ 8.0\\
Lock850.052b&             &561 $\pm$ 86&& $3.95^{\rm +2.2}_{-2.7}$ & 2.51$_{-1.02}^{\rm +1.03}$ & &Confused\\
Lock850.053 & --26 $\pm$ 20  &168 $\pm$ 15&& $4.45^{\rm +2.3}_{-2.9}$ && &$<$21\\
Lock850.060 & 4 $\pm$ 20    &87.8 $\pm$ 12.0&& $3.15^{\rm +1.7}_{-2.0}$ && &
\\
Lock850.063 & 31 $\pm$ 20   &236 $\pm$ 17&34.9 $\pm$ 29.1& $3.65^{\rm +1.2}_{-1.3}$ && &22.6 $\pm$ 4.8\\
Lock850.064 &             &425 $\pm$ 25&11.6 $\pm$ 12.4& $5.85^{\rm +2.5}_{-3.2}$ &&1.7 $\pm$ 0.4 &45.5 $\pm$ 7.4\\
Lock850.066 &             &71.2 $\pm$ 12.1&& $4.25^{\rm +1.9}_{-2.2}$ && &$<$21\\
Lock850.067 & 59 $\pm$ 20   &108 $\pm$ 14&& $2.55^{\rm +1.5}_{-1.5}$ && &$<$21\\
Lock850.070 & 79 $\pm$ 20   &106 $\pm$ 12&& $3.85^{\rm +2.2}_{-2.5}$ && &21.9 $\pm$ 7.2\\
Lock850.071 &             &181 $\pm$ 20&& $3.95^{\rm +1.8}_{-2.0}$ & 2.89$_{-1.13}^{\rm +1.10}$ & &95.8 $\pm$ 4.6\\
Lock850.073 & 89 $\pm$ 36   &278 $\pm$ 19&& $3.55^{\rm +1.9}_{-2.3}$ && &27.3 $\pm$ 4.8\\
Lock850.073b& 143 $\pm$ 61  &278 $\pm$ 19&& $3.55^{\rm +1.9}_{-2.3}$ && &26.7 $\pm$ 4.6\\
Lock850.075 &             &147 $\pm$ 17&& $4.45^{\rm +2.2}_{-2.6}$ && &
\\
Lock850.076 & 227 $\pm$ 20  &592 $\pm$ 26&4.4 $\pm$ 6.7& $4.75^{\rm +2.5}_{-3.1}$ &4.4 $\pm$ 1.4& &48.0 $\pm$ 6.0\\
Lock850.077 & 24 $\pm$ 14   &51.7 $\pm$ 13.1&41.4 $\pm$ 24.6& $3.25^{\rm +1.2}_{-1.3}$ && &15.5 $\pm$ 4.4\\
Lock850.077b& 2 $\pm$ 15    &154 $\pm$ 15&& $3.25^{\rm +1.2}_{-1.3}$ && &39.5 $\pm$ 7.8\\
Lock850.078 &             &85.6 $\pm$ 14.7&& $4.55^{\rm +2.2}_{-2.7}$ && &$<$23\\
Lock850.079 & 41 $\pm$ 15   &292 $\pm$ 18&& $3.15^{\rm +1.3}_{-1.5}$ & 2.35$_{-1.22}^{\rm +1.12}$ & &22.41 $\pm$ 4.5\\
Lock850.081 &             &3667 $\pm$ 51&& $5.35^{\rm +1.9}_{-2.3}$ && &55.2 $\pm$ 5.3\\
Lock850.083 &             &344 $\pm$ 25&& $3.15^{\rm +2.0}_{-2.1}$ && &$<$28\\
Lock850.087 &             &399 $\pm$ 22&& $3.45^{\rm +1.5}_{-1.7}$ && &84.5 $\pm$ 5.3\\
Lock850.100 &             &118 $\pm$ 13&& $11.25^{\rm +4.2}_{-5.3}$ && &19.8 $\pm$ 6.3\\[2pt]
\hline
\end{tabular}
\tablefoot {The $15~\mu$m data are presented for the first time here, and the remaining
data are taken from Coppin et~al. (\cite{Copp06}), Austermann et~al. (\cite{Aust10}), Ivison et~al. (\cite{Ivis07}), Coppin et~al. (\cite{Copp08}) and Negrello et~al. (\cite{Negr09}).}
\end{table}